Author's response to reviews

Title: Association between promoter -1607 polymorphism of MMP1 and Lumbar Disc Disease in Southern Chinese

Authors:

You-Qiang Song (songy@hku.hk)
Daniel WH Ho (dwhho@hku.hk)
Jaro Karppinen (jaro.karppinen@ttl.fi)
Patrick Y. P. Kao (patrickkao@hku.hk)
Bao-Jian Fan (bao-jian_fan@meei.harvard.edu)
Shea-Ping P Yip (hrmoldk@hku.hk)
Keith D.K. Luk (htspyip@smtp7.polyu.edu.hk)
John C.Y. Leong (jcyleong@ouhk.edu.hk)
Kathryn S.E. Cheah (hrmbdkc@hkusua.hku.hk)
Pak Sham (pcsham@hku.hk)
Danny Chan (chand@hkusa.hku.hk)
Kenneth M.C. Cheung (cheungmc@hku.hk)

Version: 3 Date: 14 March 2008

Author’s response to reviews:

Dear Editors:

Thank you for accepting our manuscript. We have changed all requested formatting.

Regards

Ken Cheung